Author’s response to reviews

Title: A rare presentation of traumatic wound on the lower lip: Case Report

Authors:

Jaume Miranda-Rius (jmiranda-rius@ub.edu)
Lluís Brunet-Llobet (lbrunet@hsjdbcn.org)
Eduard Lahor-Soler (eduardlahor@ub.edu)
Carlos Mendieta (cmendieta@ub.edu)

Version: 3
Date: 1 April 2014

Author’s response to reviews:

Dear Michael Kidd, Editor-in-Chief Journal of Medical Case Reports

Please find enclosed our manuscript, “A rare presentation of traumatic wound on the lower lip: Case Report” by Miranda-Rius et al, which would like to submit for publication as a case report in Journal of Medical Case Reports.

Explanation about why our manuscript should be published in your journal:
This case report is especially interesting because of its singularity. It may be helpful for a wide range of surgeons, dentists and even nurse practitioners. Newly qualified doctors and even specialists or generalists, at any stage of their career, could use the clinical image of the lesion for training in the differential diagnosis process of ulcerated wounds that typically appear on this facial area. To the best of our knowledge there are no previous reports at this lesion stage. Considering that it was due to a human bite it is remarkable the satisfactory result with a simple surgical procedure in spite of the elapsed time (12 days). We believe our clinical findings would appeal to the readership of Journal of Medical Case Reports.

All authors have approved the manuscript and agree with its submission to Journal of Medical case reports. We confirm that this manuscript has not been published elsewhere and is not under consideration by another journal.

We look forward to hearing from you at your earliest convenience.

Yours Sincerely,
Jaume Miranda-Rius, Lluís Brunet-Llobet, Eduard Lahor-Soler and Carlos Mendieta

PS: Please address all correspondence to: jmiranda-rius@ub.edu

Potential Peer Reviewers:
Prof. Eduard Valmaseda
Email: evalmaseda@ub.edu
Universitat de Barcelona
Prof. Declare Mushi
Email: mushideclare65@gmail.com
Kilimanjaro Christian Medical University
Dr. Juli Busquets
Email: jbusquets@csub.scs.es
Hospital Univesitari de Bellvitge. Universitat de Barcelona